Author's response to reviews

Title: Characterization of a murine model of monocrotaline pyrrole-induced acute lung injury

Authors:

Rio Dumitrascu (rio.dumitrascu@uglc.de)
Silke Koebrich (silke.koebrich@innere.med.uni-giessen.de)
Eva Dony (Eva.dony@innere.med.uni-giessen.de)
Norbert Weissmann (Norbert.Weissmann@innere.med.uni-giessen.de)
Rajkumar Savai (Rajkumar.Savai@innere.med.uni-giessen.de)
Soni S Pullamsetti (Soni.Pullamsetti@innere.med.uni-giessen.de)
Hossein Ardeschir Ghofrani (Ardeschir.Ghofrani@innere.med.uni-giessen.de)
Arun Samidurai (Arun.Samidurai@innere.med.uni-giessen.de)
Horst Traupe (Horst.Traupe@radiol.med.uni-giessen.de)
Werner Seeger (Werner.Seeger@innere.med.uni-giessen.de)
Friedrich Grimminger (Friedrich.Grimminger@innere.med.uni-giessen.de)
Ralph Schermuly (Ralph.Schermuly@innere.med.uni-giessen.de)

Version: 6 Date: 23 November 2008

Author’s response to reviews:

Dear Editor,

Please find our revised version of the manuscript entitled “Characterization of a murine model of monocrotaline pyrrole-induced acute lung injury”.


We provide here the final version including formatting changes suggested by you as follow:

1. abstract was structured into background, methods, results and conclusions
2. figure titles/legends were removed from the figures
3. we included email addresses for all authors on the title page according to provided instructions
4. the manuscript was again checked and is nor ready for publication

Yours sincerely,
Ralph Schermuly, PhD